FEATURE ARTICLE

Single-Molecule Michaelis–Menten Equations

S. C. Kou,[†] Binny J. Cherayil,^{‡,§} Wei Min,[‡] Brian P. English,[‡] and X. Sunney Xie^{*,‡}

Department of Statistics and Department of Chemistry and Chemical Biology, Harvard University, Cambridge, Massachusetts 02138

Received: March 22, 2005; In Final Form: June 12, 2005

This paper summarizes our present theoretical understanding of single-molecule kinetics associated with the Michaelis-Menten mechanism of enzymatic reactions. Single-molecule enzymatic turnover experiments typically measure the probability density f(t) of the stochastic waiting time t for individual turnovers. While f(t) can be reconciled with ensemble kinetics, it contains more information than the ensemble data; in particular, it provides crucial information on dynamic disorder, the apparent fluctuation of the catalytic rates due to the interconversion among the enzyme's conformers with different catalytic rate constants. In the presence of dynamic disorder, f(t) exhibits a highly stretched multiexponential decay at high substrate concentrations and a monoexponential decay at low substrate concentrations. We derive a single-molecule Michaelis-Menten equation for the reciprocal of the first moment of f(t), $1/\langle t \rangle$, which shows a hyperbolic dependence on the substrate concentration [S], similar to the ensemble enzymatic velocity. We prove that this single-molecule Michaelis-Menten equation holds under many conditions, in particular when the intercoversion rates among different enzyme conformers are slower than the catalytic rate. However, unlike the conventional interpretation, the apparent catalytic rate constant and the apparent Michaelis constant in this single-molecule Michaelis-Menten equation are complicated functions of the catalytic rate constants of individual conformers. We also suggest that the randomness parameter r, defined as $\langle (t - \langle t \rangle)^2 \rangle / \langle t \rangle^2$, can serve as an indicator for dynamic disorder in the catalytic step of the enzymatic reaction, as it becomes larger than unity at high substrate concentrations in the presence of dynamic disorder.

1. Introduction

The catalytic activity of enzymes has long been understood in terms of the Michaelis-Menten mechanism:¹ a substrate S binds reversibly with an enzyme E to form an enzyme-substrate complex ES that undergoes unimolecular decomposition to form a product P, regenerating the original enzyme E via E^0 .

$$E + S \stackrel{k_1}{\underset{k_{-1}}{\longleftrightarrow}} ES \stackrel{k_2}{\longrightarrow} E^0 + P \qquad E^0 \stackrel{\delta}{\rightarrow} E \qquad (1)$$

The rate of product formation v has a hyperbolic dependence on the substrate concentration [S], i.e., $v = k_2[E]_T[S]/([S] + K_M)$, where $K_M = (k_{-1} + k_2)/k_1$ and $[E]_T$ is the total enzyme concentration. This rate expression, the Michaelis–Menten equation, provides a highly satisfactory description of ensembleaveraged enzyme kinetics.

Recent advances in single-molecule spectroscopy and manipulation²⁻¹⁶ have now made it possible to study enzymatic reactions at the level of *single* molecules, thus raising the question of whether the Michaelis–Menten equation remains an adequate description of single-molecule kinetics. It is therefore of both conceptual and practical importance to understand how single molecule and ensemble kinetics are

reconciled, and what new information is available from single molecule data.

At the single-molecule level, an enzymatic reaction is a stochastic event, and a single-molecule experiment typically measures the waiting times for the completion of the enzymatic reaction. The probability density of these waiting times, f(t), can be obtained by recording the histogram of many turnovers over a long period of time. Therefore, single-molecule kinetics cannot be formulated in terms of enzyme concentrations, but must be formulated instead in terms of the probabilities for the enzyme to be in one of the possible states in the reaction pathway.¹⁷

We will show that single-molecule and steady-state ensemble kinetics are consistent, in that the reciprocal of the first moment of f(t), $1/\langle t \rangle$, has the same hyperbolic dependence on the substrate concentration as the enzymatic velocity described by the conventional Michaelis-Menten equation. However, f(t) provides much more kinetic information, such as the existence of reaction intermediates and dynamic disorder, which are often obscured by ensemble-averaged measurements. In particular, multiexponentiality in f(t) is a manifestation of dynamic disorder,^{4,11,18–34} which refers to fluctuations in the rate constants of the reaction caused by transitions among different enzyme conformers. These fluctuations can occur on a time scale comparable to or longer than that of the enzymatic reaction, so the rate of product formation is no longer governed by a single rate constant, but effectively by a distribution of rate constants.11,18,26,29

^{*} Corresponding author. E-mail: xie@chemistry.harvard.edu.

[†] Department of Statistics.

[‡] Department of Chemistry and Chemical Biology.

[§] Permanent address: Department of Inorganic & Physical Chemistry, Indian Institute of Science, Bangalore-560012, India.

S. C. Kou was born in Lanzhou, China, in 1974. He received a B.S. in mathematics from Peking University in 1997, followed by a Ph.D. in statistics from Stanford University in 2001 under the supervision of Prof. Bradley Efron. After completing the Ph.D. study, he joined Harvard University in 2001 as an Assistant Professor of Statistics, and is currently the John L. Loeb Associate Professor of the Natural Sciences. His research interests include stochastic inference in single-molecule biophysics, Monte Carlo methods, and Bayesian and non-parametric statistics.

Binny J. Cherayil was born in Kerala, India, in 1958. He received a B.Sc. from Madras University in 1978, an M.Sc. from the Indian Institute of Technology, Madras, in 1980, and a Ph.D. from the University of Chicago in 1987, where he worked under the supervision of Prof. Karl Freed. After working as a post-doctoral fellow with Prof. Harold Scheraga at Cornell University, he joined the Indian Institute of Science, Bangalore, in 1990, where he is now a Professor in the Department of Inorganic and Physical Chemistry. His research interests include the statistical mechanics of complex fluids, stochastic processes, and biophysics. He is on sabbatical with the Xie group at Harvard for the period 2004–2005.

Wei Min was born in Nanjing, China, in 1981, and received his B.S. in chemistry from Peking University in 2003. Currently he is a doctoral candidate in physical chemistry at Harvard University, and works with Prof. Sunney Xie on experimental and theoretical biophysics. His current interests include protein conformational dynamics, single-molecule enzymology, and non-equilibrium statistical mechanics in biological processes.

Brian P. English was born in Munich, Germany, in 1977 and received his B.A. from Cornell University in 2000. In 2003 he received an M.A. from Harvard University and is currently working toward a Ph.D. degree in the Department of Chemistry and Chemical Biology at Harvard University in Prof. Sunney Xie's group. His research concerns singlemolecule enzymology.

X. Sunney Xie was born in Beijing, China, in 1962. He received a B.S. in chemistry from Peking University in 1984, followed by a Ph.D. from the University of California at San Diego in 1990 working with Professor John Simon. After postdoctoral research with Prof. Graham Fleming at the University of Chicago, he joined the Environmental Molecular Sciences Laboratory at Pacific Northwest National Laboratory in 1992 where he later became a Chief Scientist. In 1999, he joined Harvard University as a Professor of Chemistry and Chemical Biology. His research interests are single-molecule enzymology, imaging gene expression, and coherent anti-Stokes Raman scattering microscopy.

A theory of the waiting time distribution f(t) should account for two sets of single-molecule experimental results. The first is the initial rise and subsequent decay of f(t) observed in experiments carried out by Lu et al.,⁴ Asbury et al.,³⁵ and Yasuda et al.⁷ Such rise and decay of f(t) is generally attributed to the formation of one or more intermediates and is often characterized by the randomness parameter r introduced by Block and co-workers^{36,37} as a measure of the relative magnitudes of the variance and the mean of the waiting time, $r \equiv \langle (t - \langle t \rangle)^2 \rangle / \langle t \rangle^2$. In the absence of dynamic disorder, it has been shown^{36,37} that if the reaction has only one rate-limiting step, r = 1, whereas if the reaction has more than one rate-limiting step, r < 1.

The second result is the observation made by English et al.¹⁶ that f(t) is a highly stretched multiexponential decay at high substrate concentrations and a monoexponential decay at low substrate concentrations. The nonexponential decay of f(t) is generally attributed to dynamic disorder.^{4,11,18–34} Furthermore, Lu et al.,⁴ Velonia et al.,¹⁴ and English et al.¹⁶ have observed dynamical correlations between successive enzymatic turnover events. Such memory effects are generally associated with slow conformational fluctuations of the enzyme during the course of the experiment.

In this article, we present models based on the Michaelis– Menten mechanism to account for these experimental observations.

Section 2 discusses the single-molecule Michaelis–Menten equation in the absence of dynamic disorder starting from the differential equations that define both the ensemble-averaged and single-molecule Michaelis–Menten kinetics. The conventional Michaelis–Menten equation is obtained from these equations by assuming a steady-state condition.³⁸ The corresponding single molecule differential equations, on the other hand, can be solved exactly for f(t) without making this assumption. f(t) itself exhibits a rise and decay due to the formation of an enzyme–substrate complex. The steady state condition actually corresponds to a very fast initial rise of f(t). The substrate concentration dependence of the enzymatic rate $1/\langle t \rangle$, calculated from f(t), obeys the Michaelis–Menten like equation.

Section 3 discusses the single-molecule Michaelis-Menten equation in the presence of dynamic disorder. We first consider the simplest case, in which each of the enzyme species in the reaction (E, ES, and E^0) exists in two interconverting conformers with different catalytic rate constants. Expressions for f(t) and $1/\langle t \rangle$ are derived. Under the condition of slow interconversion between the conformers, the dependence of $1/\langle t \rangle$ on [S] is again found to be identical to the ensemble Michaelis-Menten equation, except that the apparent k_2 and K_M of the singlemolecule Michaelis-Menten equation have meanings different from their conventional interpretations. Consistent with experimental findings, ${}^{16} f(t)$ changes from a highly stretched multiexponential decay to a monoexponential decay as substrate binding becomes rate limiting. As a generalization of this model, we also consider the physically more realistic case of an arbitrary number of interconverting conformers, which leads to substantially the same conclusions as the two-state model. The slow interconversion among conformers results in the memory effect associated with the correlations between successive enzymatic turnover times.

Section 4 introduces a semi-Markovian (or memoryless) approximation to the kinetic scheme of the previous section in which the catalytic step is assumed to be non-Poissonian with a general multiexponential waiting time distribution. Again, we arrive at the important conclusion that $1/\langle t \rangle$ obeys the Michaelis–Menten equation. We also find, as before, that the kinetic parameters of the single-molecule Michaelis–Menten equation (corresponding to k_2 and K_M in ensemble measurements) have meanings different from their conventional interpretations.

Section 5 discusses the substrate concentration dependence of the randomness parameter r (which is related to the second moment of f(t)), with and without dynamic disorder, the treatment of dynamic disorder following the approach discussed in section 3. While it is known that r can be less than unity because of the existence of more than one rate-limiting step,^{36,37} we show that r can also be larger than unity because of dynamic disorder. Thus, r can potentially serve as an indicator of dynamic disorder.

A summary of the main results is presented in the final section of the paper. Relevant mathematical details of the calculations are provided in the Appendices.

2. Single-Molecule Michaelis-Menten Kinetics in the Absence of Dynamic Disorder

The Michaelis–Menten mechanism for the enzymatic conversion of substrate S to product P by enzyme E is described in eq 1. The rate equations for the concentrations of the chemical species in the first reaction are therefore given by

$$\frac{d[E]}{dt} = -k_1[E][S] + k_{-1}[ES]$$
(2)

$$\frac{d[ES]}{dt} = k_1[E][S] - (k_{-1} + k_2)[ES]$$
(3)

$$\frac{\mathbf{d}[\mathbf{E}^0]}{\mathbf{d}t} = \frac{\mathbf{d}[\mathbf{P}]}{\mathbf{d}t} = k_2[\mathbf{ES}] \tag{4}$$

where *t* is the elapsed time from the onset of an ensembleaveraged experiment. The initial conditions are [ES] = 0 and $[E^0] = 0$ at t = 0. At early times, when very little substrate has been converted to product, the second reaction in eq 1 can be neglected. Because both [E] and [S] are time dependent, eqs 2-4 are nonlinear differential equations and cannot be solved exactly. However, an approximate solution for v = d[P]/dt can be obtained if the concentration of the complex, [ES], is assumed to reach a steady state shortly after the onset of the reaction. This steady-state approximation corresponds to the condition d[ES]/dt = 0, and its application to eqs 2-4 is easily shown to lead to the classic Michaelis-Menten equation³⁸

$$v = \frac{v_{\max}[S]}{[S] + K_{M}}$$
(5)

where v_{max} , defined as $v_{\text{max}} = k_2[E]_T$, with $[E]_T = [E] + [ES]$ the total enzyme concentration, is the reaction velocity at saturating substrate concentration, and K_M , the Michaelis constant, defined as $K_M = (k_{-1} + k_2)/k_1$, is the substrate concentration at which the enzymatic velocity is half of v_{max} .

In a turnover experiment, a single enzyme molecule is monitored continuously as it cycles repetitively through the states E, ES, and E^0 in eq 1. The time for the first reaction to complete is now a stochastic variable that can be completely characterized by a waiting time distribution f(t). To derive the rate equations that describe the corresponding single-molecule Michaelis—Menten kinetics, the concentrations in eqs 2–4 are replaced by the probabilities P of finding the single enzyme molecule in the states E, ES, and E^0 , leading to the equations

$$\frac{\mathrm{d}P_{\rm E}(t)}{\mathrm{d}t} = -k_1^{\ 0}P_{\rm E}(t) + k_{-1}P_{\rm ES}(t) \tag{6}$$

$$\frac{\mathrm{d}P_{\mathrm{ES}}(t)}{\mathrm{d}t} = k_1^{\ 0} P_{\mathrm{E}}(t) - (k_{-1} + k_2) P_{\mathrm{ES}}(t) \tag{7}$$

$$\frac{\mathrm{d}P_{\mathrm{E}^{0}}(t)}{\mathrm{d}t} = k_{2}P_{\mathrm{ES}}(t) \tag{8}$$

which must satisfy the initial conditions $P_{\rm E}(0) = 1$, $P_{\rm ES}(0) = 0$, and $P_{\rm E^0}(0) = 0$ at t = 0 (the time of onset of the reaction), along with the constraint $P_{\rm E}(t) + P_{\rm ES}(t) + P_{\rm E^0}(t) = 1$. Also, the rate constant for the forward step, k_1^0 , is treated as a pseudo-first-order rate constant that can be written as $k_1^0 = k_1[{\rm S}]$, with [S] assumed to be time-independent. This is reasonable, as there is essentially no depletion of substrate by a single enzyme molecule, and [S] can be considered as a constant. ${\rm E}^0$ is converted back to E through the second half reaction in eq 1. Depending on the enzyme system, this can occur either instantaneously (E and E⁰, thereby becoming effectively identical¹⁶), or through another chemical reaction via the ping-pong mechanism.⁴



Figure 1. Probability density of the waiting time, f(t), in the absence of dynamic disorder, as calculated from eq 10, for three different values of k_{-1} (0, 50, and 2000 s⁻¹) with $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, $k_2 = 250 \text{ s}^{-1}$, and [S] = 0.005 mM.

Consequently, eqs 6–8 become a system of *linear* first-order differential equations, and they can be solved exactly for $P_{\rm E}(t)$, $P_{\rm ES}(t)$, and $P_{\rm E^0}(t)$. Knowing $P_{\rm E^0}(t)$, the waiting time distribution f(t), which is normalized such that $\int_0^{\infty} dt f(t) = 1$, is obtained as follows: The probability that a turnover occurs between t and $t + \Delta t$ is $f(t)\Delta t$; $f(t)\Delta t$ is the same as the probability that the enzyme is in the state ${\rm E}^0$ in the interval between t and $t + \Delta t$, which is $\Delta P_{\rm E^0}(t) = k_2 P_{\rm ES}(t)\Delta t$. Thus, in the limit of infinitesimal Δt ,

$$f(t) = dP_{E0}(t)/dt = k_2 P_{ES}(t)$$
(9)

From the solutions of eqs 6–8, and using the above relation for f(t), it is easily shown that⁴

$$f(t) = \frac{k_1 k_2 [S]}{2A} [\exp(A + B)t - \exp(B - A)t]$$
(10)

where $A = \sqrt{(k_1[S] + k_{-1} + k_2)^2/4 - k_1k_2[S]}$ and $B = -(k_1[S] + k_{-1} + k_2)/2$, and the substrate concentration dependence [S] has been shown explicitly through the relation $k_1^0 = k_1[S]$.

A plot of f(t) vs t at fixed values of [S] (0.005 mM), k_1 (10⁷ M⁻¹ s⁻¹), and k_2 (250 s⁻¹) is shown in Figure 1 for three different values of k_{-1} (0, 50, and 2000 s⁻¹.) These values of k_{-1} are illustrative of reactions in which (i) the dissociation of ES to E and S does not occur (top panel), (ii) the catalytic and dissociation rates are roughly comparable (middle panel), and (iii) the rate of dissociation of ES to E and S is significantly larger than the catalytic rate, leading to steady-state formation of ES (bottom panel).

The limit $k_{-1} \rightarrow 0$ of the top panel describes the sequential reaction $S + E \xrightarrow{k_1^0} ES \xrightarrow{k_2} E^0 + P$. The waiting time distribution f(t) of such a reaction is the convolution of the waiting time distributions $f_1(t)$ and $f_2(t)$ of the two separate steps, i.e., f(t) = $(f_1 \otimes f_2)(t)$, or $f(t) = \int_0^t dt' f_1(t - t')f_2(t')$. If $f_1(t)$ and $f_2(t)$ are $k_1[S] \exp(-k_1[S]t)$ and $k_2 \exp(-k_2t)$, respectively, (implying that the steps $E + S \rightarrow ES$ and $ES \rightarrow E^0 + P$ are Poisson processes), then f(t) is given exactly by

$$f(t) = \frac{k_1 k_2[\mathbf{S}]}{k_2 - k_1[\mathbf{S}]} \left(\exp(-k_1[\mathbf{S}]t) - \exp(-k_2t) \right) \quad (11)$$



Figure 2. Probability density of the waiting time, f(t), in the absence of dynamic disorder, as calculated from eq 10, for three different values of [S] (0.020 mM, 0.010 mM, and 0.005 mM) with $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, $k_2 = 250 \text{ s}^{-1}$, and $k_{-1} = 50 \text{ s}^{-1}$, respectively.



Figure 3. Average reaction rate $1/\langle t \rangle$ or its equivalent $\nu/[E]_T$, as calculated from eq 14b, the single-molecule Michaelis–Menten equation, as a function of substrate concentration [S] for $K_M = 30 \ \mu$ M (the value corresponding to $k_{-1} = 50 \ s^{-1}$, $k_2 = 250 \ s^{-1}$, and $k_1 = 10^7 \ M^{-1} \ s^{-1}$).

and exhibits an exponential rise followed by an exponential decay, corresponding to the generation of the intermediate ES, with the faster of k_1 and k_2 being the rate constant of the rise, and the slower of k_1 and k_2 being the rate constant of the decay.¹⁷

Another limit of f(t), shown in the bottom panel, exhibits only a single-exponential decay and corresponds to the steady-state limit in which ES is generated essentially immediately. In analogy with the ensemble steady-state approximation, this limit can be expressed analytically as $dP_{\rm ES}(t)/dt = 0$, and typically holds when $k_2 \ll k_{-1}$. Combined with the constraint $P_{\rm E}(t) + P_{\rm ES}(t) + P_{\rm E^0}(t) = 1$ and the initial condition $P_{\rm E^0}(0) = 0$, the steady-state limit $dP_{\rm ES}(t)/dt = 0$ applied to eqs 6–8 can be shown to lead to

$$P_{\rm E^0}(t) = 1 - \exp\left[-\frac{k_1^0 k_2 t}{k_1^0 + k_{-1} + k_2}\right]$$
(12)

which, using eq 9, leads in turn to

$$f(t) = \frac{k_1 k_2[\mathbf{S}]}{k_1[\mathbf{S}] + k_{-1} + k_2} \exp\left[-\frac{k_1 k_2[\mathbf{S}]t}{k_1[\mathbf{S}] + k_{-1} + k_2}\right] (13)$$

confirming the single-exponential decay of f(t). Furthermore, in the limit of high concentration, f(t) reduces to $k_2 \exp(-k_2 t)$, as expected.

The middle panel describes an intermediate case between the two limits described above. At such an intermediate value of k_{-1} , the dependence of f(t) on [S] is illustrated in Figure 2 for fixed values k_1 , k_{-1} , and k_2 (10⁷ M⁻¹ s⁻¹, 50 s⁻¹, and 250 s⁻¹, respectively.)

The first moment of f(t), $\langle t \rangle = \int_0^{\infty} dt t f(t)$, gives the mean waiting time $\langle t \rangle$ for the reaction, from which the connection with the ensemble measurements under steady-state conditions can be made. The reciprocal of $\langle t \rangle$ can be interpreted as an average reaction rate.^{39,40} Generally, this arises from the equivalence between time averaging and ensemble averaging. From eq 10, we deduce that

$$\frac{1}{\langle t \rangle} = -\frac{(A^2 - B^2)^2}{2Bk_1k_2[S]}$$
(14a)

$$=\frac{k_2[S]}{[S]+K_{\rm M}}\tag{14b}$$

A comparison of eqs 5 and 14b indicates that $v/v_{\text{max}} = 1/(k_2 \langle t \rangle)$ or that $v/[E]_{\text{T}} = 1/\langle t \rangle$.

This is a gratifying result, indicating that the first moment of f(t) does indeed recover the classic Michaelis—Menten equation, regardless of whether the steady-state approximation is used in the single-molecule probability calculation. We regard eq 14b as the single-molecule Michaelis—Menten equation.

A plot of $1/\langle t \rangle$ against [S] for the parameters used in the middle panel of Figure 1 is shown in Figure 3, exhibiting the characteristic hyperbolic profile of the classic Michaelis– Menten saturation curve. The fact that $1/\langle t \rangle$ calculated from eq 14b exactly coincides with eq 5 highlights the consistency between the single-molecule and ensemble-averaged kinetics. However, it is important to stress that f(t) does provide more information than only the first moment, such as higher moments and the existence of intermediates. This is particularly true in the presence of dynamic disorder, as will be discussed next.

3. Single-Molecule Michaelis-Menten Kinetics in the Presence of Dynamic Disorder

The expressions for f(t) derived in section 2 are not consistent with measurements on some enzyme systems,^{14–16} which show significant multiexponentiality in the waiting time distribution at high substrate concentrations. This behavior can be attributed to dynamic disorder. One way to model dynamic disorder is to assume, as in the approach used by Zwanzig²⁰ and by Yang and Cao,²⁸ that the rate constant k_2 , or the parameters on which it depends, are stochastic variables that fluctuate according to some prescribed statistics. However, the main goal of the rest of the paper is not to provide specific models for the fluctuations of these stochastic variables, but to explain the multiexponentiality of f(t) and its concentration dependence, and to establish the general applicability of the Michaelis—Menten equation to single-molecule kinetics even in the presence of dynamic disorder.

(i) **Two-State Model.** To this end, we first consider the simplest extension of the Michaelis–Menten mechanism that incorporates the notion of dynamic disorder. This is the kinetic scheme in which the three states of the enzyme, E, ES, and E⁰,

can each exist in two interconverting conformations, as shown below:

$$S + E_{1} \xrightarrow[k_{-11}]{k_{-11}} ES_{1} \xrightarrow{k_{21}} P + E_{1}^{0} \xrightarrow{\delta_{1}} E_{1}$$

$$\alpha \not \uparrow \alpha \qquad \beta \not \uparrow \beta \qquad \gamma \not \downarrow \uparrow \gamma \qquad (15)$$

$$S + E_{2} \xrightarrow[k_{-12}]{k_{-12}} ES_{2} \xrightarrow{k_{22}} P + E_{2}^{0} \xrightarrow{\delta_{2}} E_{2}$$

Even this simple generalization presents theoretical challenges in the calculation of f(t) and its first moment. An immediate complication is that over the course of a long time trajectory, each new reaction cycle begins from either E_1 or E_2 with a probability that reflects the steady-state populations of the various intermediates. This means that f(t) must be calculated from the weighted average

$$f(t) = w_1 f_{T_{E_1}}(t) + w_2 f_{T_{E_2}}(t)$$
(16)

where $f_{T_{E_1}}(t)$ and $f_{T_{E_2}}(t)$ are the distributions of the waiting times T_{E_1} and T_{E_2} for the enzyme to complete the reaction starting from E_1 and E_2 , respectively, and w_1 and w_2 are the corresponding steady-state probabilities for the enzyme to exist in one or other of these conformations. We find that while w_1 and w_2 can be calculated from the master equation formalism used to derive eqs 6–8, the calculation of f(t) requires a different approach and can only be found in closed form in the Laplace domain. Details of the complete calculation are provided in Appendix A; here we state only the final result, which can be written as

$$\hat{f}(s) = (w_1, w_2, 0, 0)\hat{\mathbf{f}}(s)$$
 (17)

where $\hat{f}(s)$ is the Laplace transform of f(t), and $\hat{\mathbf{f}}(s) \equiv (\hat{f}_{T_{E_1}}(s), \hat{f}_{T_{E_3}}(s), \hat{f}_{T_{E_3}}(s))^{\mathrm{T}}$ is defined by

$$\hat{\mathbf{f}}(s) = (s\mathbf{I} - \mathbf{Q})^{-1}\mathbf{r}$$
(18)

where **I** is the identity matrix, $\mathbf{r} \equiv (0, 0, k_{21}, k_{22})^{\mathrm{T}}$, $f_{T_{\mathrm{ES}_1}}(t)$ and $f_{T_{\mathrm{ES}_2}}(t)$ are the distributions of the waiting times T_{ES_1} and T_{ES_2} for the enzyme to complete the reaction starting from ES₁ and ES₂, respectively, and

$$\mathbf{Q} = \begin{pmatrix} -(\alpha + k_{11}[\mathbf{S}]) & \alpha & k_{11}[\mathbf{S}] & 0 \\ \alpha & -(\alpha + k_{12}[\mathbf{S}]) & 0 & k_{12}[\mathbf{S}] \\ k_{-11} & 0 & -(\beta + k_{-11} + k_{21}) & \beta \\ 0 & k_{-12} & \beta & -(\beta + k_{-12} + k_{22}) \\ \end{pmatrix}$$
(19)

In the limit of the fast reset of E_1^0 and E_2^0 to E_1 and E_2 , corresponding to the condition δ_1 , $\delta_2 \gg 1$, the steady-state weights w_1 and w_2 , which satisfy

$$w_1 + w_2 = 1 \tag{20}$$

can be found from

$$\frac{w_1}{w_2} = \frac{k_{21}[\alpha(k_{11}k_{22} + k_{11}k_{-12}) + \alpha\beta(k_{11} + k_{12}) + \beta k_{11}k_{12}[S]]}{k_{22}[\alpha(k_{12}k_{21} + k_{12}k_{-11}) + \alpha\beta(k_{11} + k_{12}) + \beta k_{11}k_{12}[S]]}$$
(21)

Equations 16–21 provide the complete solution in Laplace space to the waiting time distribution of the two-state model of dynamic disorder. The first moment of f(t), $\langle t \rangle$, is easily obtained from the formula $\langle t \rangle = -d\hat{f}(s)/ds|_{s=0}$. After lengthy but straightforward algebra, one can show that

$$\frac{1}{\langle t \rangle} = \frac{F^{-1}[S]}{[S] + \frac{G[S] + H}{FJ[S] + FK}}$$
(22)

where the constants F, G, H, and K are given by

$$F = \frac{2}{k_{21} + k_{22}} \tag{23a}$$

$$G = \alpha(k_{21} - k_{22}) \frac{k_{12}(k_{21} + k_{-11}) - k_{11}(k_{22} + k_{-12})}{k_{21} + k_{22}} + \beta[k_{12}(k_{21} + k_{-11}) + k_{11}(k_{22} + k_{-12})]$$
(23b)

$$H = 2\alpha(k_{22} + k_{-12})(k_{21} + k_{-11}) + 2\alpha\beta(k_{21} + k_{22} + k_{-11} + k_{-12})$$
(23c)

$$J = \beta k_{11} k_{12} (k_{21} + k_{22}) \tag{23d}$$

$$K = \alpha[k_{11}k_{21}(k_{22} + k_{-12}) + k_{12}k_{22}(k_{21} + k_{-11})] + \alpha\beta(k_{11} + k_{12})(k_{21} + k_{22})$$
(23e)

Interpreting $1/\langle t \rangle$ as the ensemble rate (by the assumption of ergodicity), one sees from eq 22 that this rate does not always obey the Michaelis-Menten equation, which is characteristically hyperbolic in the substrate concentration [S]. However, there are a number of limiting conditions that do produce this hyperbolic relationship. In particular, a Michaelis-Menten-like equation is recovered if one of the following conditions (a-f) holds: (a) $k_{21} \gg \beta$, $k_{22} \gg \beta$, corresponding to the limit in which the catalytic rates k_{21} and k_{22} are much larger than the interconversion rate β between ES₁ and ES₂; (b) $\beta \rightarrow 0$, corresponding to the limit of slow interconversion between ES₁ and ES₂; (c) $\alpha \rightarrow 0$, corresponding to the limit of slow interconversion between E_1 and E_2 ; (d) $\alpha \rightarrow \infty$, corresponding to the limit of fast interconversion between E_1 and E_2 ; (e) $(k_{21} + k_{-11})/k_{11} = (k_{22} + k_{-22})/k_{12}$, corresponding to the case where the two channels $E_1 + S \nleftrightarrow ES_1 \rightarrow E_1^0 + P$ and $E_2 + S \leftrightarrow ES_2 \rightarrow E_2^0 + P$ have identical Michaelis constants; and (f) $\beta \rightarrow \infty$ and $k_{11} = k_{12}$, corresponding to the limit of fast interconversion between ES1 and ES2, and an identical rate constant of interconversion for the steps E_1 to ES_1 and E_2 to ES₂.

Condition (a) is not very stringent, especially in light of recent observations of slow conformational fluctuations.^{12,34} If after imposing this condition, one also takes α , the interconversion rate constant between E₁ and E₂, to be small, the disorder is effectively quasi-static. In this quasi-static disorder limit, there is a time scale separation between the fast catalytic reaction and the sluggish interconversions between the conformers of the enzyme and the enzyme—substrate complex. In this limit, it can be shown (see Appendix B) that the steady-state waiting time distribution is well approximated by

$$f(t) = \sum_{i=1}^{2} w_i \frac{k_{1i} k_{2i} [S]}{2A_i} [\exp(A_i + B_i)t - \exp(B_i - A_i)t] \quad (24)$$

where $A_i = \sqrt{(k_{1i}[S] + k_{-1i} + k_{2i})^2 / 4 - k_{1i} k_{2i}[S]}, B_i = -(k_{1i}[S] + k_{-1i} + k_{2i}) / 2$, and the weights w_1 and w_2 are

$$w_{1} = \frac{k_{11}k_{21}(k_{22} + k_{-12})}{k_{11}k_{21}(k_{22} + k_{-12}) + k_{12}k_{22}(k_{21} + k_{-11})}$$
$$w_{2} = \frac{k_{12}k_{22}(k_{21} + k_{-11})}{k_{11}k_{21}(k_{22} + k_{-12}) + k_{12}k_{22}(k_{21} + k_{-11})}$$
(25)

If $1/\langle t \rangle$ is calculated from eq 24, we arrive at the single-molecule Michaelis–Menten equation for the two conformer case,

$$\frac{1}{\langle t \rangle} = \frac{\chi_2'[S]}{[S] + C_M'}$$
(26)

where the apparent catalytic rate constant χ'_2 and the apparent Michaelis constant C'_M , unlike k_2 and K_M in eq 5, are found to be

$$1/\chi_2' = w_1/k_{21} + w_2/k_{22} \tag{27a}$$

$$C'_{\rm M} = \chi'_2[w_1 K_{\rm M1}/k_{21} + w_2 K_{\rm M2}/k_{22}]$$
 (27b)

with $K_{Mi} \equiv (k_{-1i} + k_{2i})/k_{1i}$. χ'_2 is nothing but the weighted harmonic mean of the catalytic rate constants in the two channels, while C'_M is a more complex function of the catalytic and Michaelis constants of the two conformers.

The significance of eq 26 and eq 27 is that single molecule Michaelis—Menten equation holds even under the condition of dynamic disorder, though χ'_2 and C'_M have different meanings from k_2 and K_M in the conventional Michaelis—Menten equation. We note that at the *ensemble* level, kinetic schemes involving multiple states similar to eq 15 have been shown³⁸ to lead to the Michaelis—Menten equation with redefined k_2 and K_M . However, we will show below that this is true for an arbitrary number of conformers.

(ii) Multistate Model. The two-state model of the foregoing section illustrates the effects on single molecule enzymatic trajectories of fluctuations between *pairs* of conformers, but real enzyme systems are likely to interconvert among a much larger number of conformational substates.^{12,16,33,34} In this section we therefore consider a generalization of the two-state model in which each of the enzyme species in eq 1, E, ES, and E⁰, is allowed to exist in any number *n* of mutually interconverting conformers. This *n*-state model of the Michaelis–Menten mechanism leads to the kinetic scheme shown below:

$$S + E_{1} \xrightarrow{k_{11}[S]} ES_{1} \xrightarrow{k_{21}} P + E_{1}^{0} \xrightarrow{\delta_{1}} E_{1}$$

$$\alpha_{12} \not \uparrow \alpha_{21} \qquad \beta_{12} \not \uparrow \beta_{21} \qquad \gamma_{12} \not \uparrow \gamma_{21}$$

$$S + E_{2} \xrightarrow{k_{12}[S]} ES_{2} \xrightarrow{k_{22}} P + E_{2}^{0} \xrightarrow{\delta_{2}} E_{2}$$

$$\vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots \qquad \vdots$$

$$A^{\dagger} \qquad \downarrow \uparrow \qquad \downarrow \uparrow \qquad \downarrow \uparrow$$

$$S + E_{n} \xrightarrow{k_{1n}[S]} ES_{n} \xrightarrow{k_{2n}} P + E_{n}^{0} \xrightarrow{\delta_{n}} E_{n}$$

$$(28)$$

In this scheme, it should be understood that E_i not only interconverts with E_{i+1} or E_{i-1} , but does so with all the other conformers as well. And the same is true for the conformer

ES_{*i*}. The calculation of the waiting time distribution for this scheme follows exactly the approach used earlier, except that many of the steps must be reformulated in terms of matrices. We first calculate the waiting time distributions through different channels; we then determine their steady-state average to obtain the overall waiting time distribution f(t), the experimentally observed quantity. The details of the calculation are lengthy, and are provided in Appendix B. Here we point out only that when the reset of E_1^{0} , E_2^{0} , ... to E_1 , E_2 , ... is much faster than any of the other interconversion steps, it can be shown that under physically meaningful conditions the average enzymatic rate, $1/\langle t \rangle$, again follows a Michaelis–Menten-like equation in which the apparent catalytic rate constant and the apparent Michaelis constant are complicated functions of the various interconversion rates (see Appendix B).

The conditions leading to the Michaelis–Menten form include: (a) the limit in which the catalytic rates are much greater than the interconversion rates of the enzyme–substrate complex ES_i, (b) the limit of extremely slow interconversion between the enzyme–substrate complexes ES_i, (c) the limit of extremely slow interconversion between the enzymes E_i, (d) the limit in which interconversion rates between the E_i's are much greater than all the other rates, (e) the limit in which the Michaelis constants for a given reaction channel are nearly the same, $(k_{21} + k_{-11})/k_{11} = (k_{22} + k_{-22})/k_{12} = ... = (k_{2n} + k_{-1n})/k_{1n}$, and the interconversion rates between the different conformers are symmetric: $\alpha_{ij} = \alpha_{ji}$, $\beta_{ij} = \beta_{ji}$, and (f) the limit in which the interconversion rates β_{ij} and β_{ji} are equal, and are much faster than the other rates.

As in the two-state model, condition (a) is of direct relevance to real enzyme systems. If, after imposing this condition, the interconversion rates between the conformers of the enzyme are also made small, the disorder is effectively quasi-static, and as shown in Appendix B, the waiting time distribution f(t) is then well approximated by

$$f(t) = \frac{1}{\sum_{i=1}^{n} w_i} \sum_{i=1}^{n} w_i \frac{k_{1i}k_{2i}[\mathbf{S}]}{2A_i} [\exp(A_i + B_i)t - \exp(B_i - A_i)t]$$
(29)

where the w_i are the steady-state weights with which each reaction channel *i* contributes to the overall waiting time distribution, and the parameters A_i and B_i have the same definitions as the corresponding parameters in eq 24.

The use of eq 29 to calculate the mean enzymatic rate leads once again to the single-molecule Michaelis-Menten equation, in the form

$$\frac{1}{\langle t \rangle} = \frac{\chi_2[S]}{[S] + C_{\rm M}} \tag{30}$$

where the apparent catalytic rate constant χ_2 and the apparent Michaelis constant C_M can be written as

$$1/\chi_2 = \sum_{i=1}^n w_i / k_{2i}$$
(30a)

$$C_{\rm M} = \chi_2 \sum_{i=1}^n w_i K_{{\rm M}i} / k_{2i} = \chi_2 \sum_{i=1}^n w_i (k_{-1i} + k_{2i}) / (k_{1i} k_{2i})$$
(30b)

Thus, χ_2 and C_M have the same structure as the corresponding kinetic constants for the two-state model; i.e., χ_2 is the



Figure 4. Probability density of the waiting time, f(t), in the presence of dynamic disorder for three different concentrations (10, 20, and 100 μ M), as calculated from eq 31, with $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, $k_{-1} = 50 \text{ s}^{-1}$, and using the parameter values a = 6, b = 35 in the gamma distribution $w(k_2) = [1/b^a \Gamma(a)] k_2^{a-1} \exp(-k_2/b)$.

weighted harmonic mean of the catalytic rate constants along the individual reaction channels, and $C_{\rm M}$ is a more complex function of the interconversion rates.

It is also interesting to note that the weights w_i can themselves be expressed in terms of the catalytic efficiency of individual conformers. The catalytic efficiency K_{Ei} of an individual conformer is defined as the ratio of its catalytic rate constant k_{2i} to its Michaelis constant K_{Mi} , i.e., $K_{Ei} = k_{2i}/K_{Mi} = k_{2i}k_{1i}/(k_{-1i} + k_{2i})$; w_i can be shown to be $w_i = K_{Ei}/\sum_{i=1}^{n} K_{Ei}$.

For the purpose of comparison with experiment, it is convenient to simplify eq 29 further by assuming that $k_{11} = k_{12}$ $= \dots = k_{1n} \equiv k_1$, and that $k_{-11} = k_{-12} = \dots = k_{-1n} \equiv k_{-1}$. Additionally, if *n* is large (as is generally the case), a continuum approximation can be invoked. These simplifications then lead to

$$f(t) = \int_0^\infty dk_2 \, w(k_2) \frac{k_1 k_2 [S]}{2A} [\exp(A+B)t - \exp(B-A)t]$$
(31)

where *A* and *B* are identical to the corresponding expressions that appear in eq 10. It is reasonable to further assume that the weight function $w(k_2)$ is a gamma distribution, such that $w(k_2) = [1/b^a \Gamma(a)]k_2^{a-1} \exp(-k_2/b)$, *a* and *b* being adjustable parameters. With this choice of weight function, the integral in eq 31 can be evaluated exactly. The resulting f(t) is shown in Figure 4 as a function of *t* at different [S] for the following arbitrary parameter values: a = 6, b = 35, $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, and $k_{-1} = 50 \text{ s}^{-1}$. The curves clearly illustrate how, as the substrate concentration increases, f(t) increasingly departs from single-exponential decay behavior.

These trends are in complete qualitative agreement with experimental results,¹⁶ and they may be explained as follows. At low substrate concentrations, the binding of the enzyme to the substrate is the rate-limiting step in the reaction, so f(t) reflects the statistics of this Poissonian step, which is therefore governed by an exponential distribution. At high substrate concentrations, the dissociation of the enzyme—substrate complex to product is the rate-limiting step in the reaction, so f(t) now reflects the statistics of this step, which is no longer Poissonian (because of dynamic disorder).

The calculation of $1/\langle t \rangle$ from eq 31 using the given expression for $w(k_2)$ readily establishes that $\chi_2 = (a - 1)b$ and $C_M =$



Figure 5. Substrate concentration dependence of the mean enzymatic velocity $1/\langle t \rangle$, or its equivalent $\nu/[E]_T$, according to the single-molecule Michaelis–Menten equation (eq 30), under the condition of the slow interconversion among conformers of a broad distribution of k_2 . The parameters are the same as described in the caption of Figure 4.

 $(k_{-1} + \chi_2)/k_1$. Figure 5, showing the variation of $1/\langle t \rangle$ with [S] for the same set of parameters used in Figure 4, is characteristically hyperbolic, as is the ensemble enzymatic velocity.

Since the mean of the gamma distribution, k_2 , is ab, and the variance, Δ , is ab^2 , the apparent catalytic rate constant can be written $\chi_2 = \bar{k}_2 - \Delta/\bar{k}_2$. Assuming a = 6, b = 35, if \bar{k}_2 is kept constant at 210 s⁻¹ and Δ is increased by a factor of 4, χ_2 decreases from 175 s⁻¹ to 70 s⁻¹. This result has interesting implications for the interpretation of different apparent catalytic constants, χ_2 . A decrease in χ_2 need not be associated with an overall decrease in the mean catalytic constant, \bar{k}_2 , but could arise from a larger variance, Δ .

The single-molecule Michaelis-Menten equation for the multiple conformer case, eq 30, explains why the conventional Michaelis-Menten equation, eq 5, is so widely applicable, since even in the presence of dynamic disorder for each single molecule, the hyperbolic concentration dependence of $1/\langle t \rangle$ almost invariably holds. In the presence of dynamic disorder, however, the constants k_2 and K_M in the ensemble Michaelis-Menten equation must be reinterpreted. In the slow interconversion limit, they are now seen to be weighted averages of the kinetic parameters characterizing individual conformers. Thus, the experimental observation of a hyperbolic dependence of the enzyme velocity on the substrate concentration does not imply that eq 1 accurately describes the underlying kinetic scheme, since a more complicated scheme, such as eq 28 can produce seemingly identical results. Pre-steady-state ensemble-averaged measurements can, in principle, distinguish the dispersed kinetics. However, in practice, they often do not have high enough dynamic range for accurate determination of multiexponential kinetics. We demonstrate that single-molecule measurements of f(t) provide a sensitive measure of dynamic disorder.

4. Semi-Markovian (Memoryless) Approximation to Multistate Model

The model of single molecule kinetics based on conformational fluctuations introduced in the previous section provides detailed microscopic interpretations of ensemble rate expressions. However, these expressions [cf. Appendix B] are quite complex, so it is worthwhile to consider alternatives that capture key experimental observations without being algebraically complex. This section introduces a semi-Markovian approximation to the multistate model that leads to simpler expressions for both the ensemble reaction rate and the waiting time distribution; but the approximation, by its Markovian nature, does not account for the memory effect. In this approximation, the distinct conformational states E_i , ES_i , and E_i^0 in the multistate model are collapsed into a set of effective states E, ES, and \mathbf{E}^{0} , and the corresponding reaction mechanism can then be written as $\mathbf{E} + \mathbf{S} \leftrightarrow \mathbf{ES} \rightarrow \mathbf{E}^0 + \mathbf{P}$. In this reduced description, the binding step $\mathbf{E} + \mathbf{S} \rightarrow \mathbf{ES}$ and the dissociation step $\mathbf{ES} \rightarrow$ $\mathbf{E} + \mathbf{S}$ are assumed to be governed by monoexponential distributions with rate constants $k_1[S]$ and k_{-1} respectively. However, the catalytic step of the reaction, $\mathbf{ES} \rightarrow \mathbf{E}^0 + \mathbf{P}$, is assumed to be governed not by a single well-defined rate constant k_2 but by a waiting time distribution $f_{T_c}(t)$, which can be specified arbitrarily. We shall refer to this scheme as a semi-Markovian approximation.⁴¹ (In the context of the kinetic scheme described by eq 28, this semi-Markovian approximation becomes exact if the interconversion rate constants β_{ii} for the complexes ES_i are much larger than the catalytic rate constants k_{2i} , and the interconversion rate constants α_{ii} for the enzyme E_i approach infinity.)

The overall waiting time distribution f(t) in this picture is now a function of $f_{T_{C}}(t)$ and is shown in Appendix C to be given by

$$\hat{f}(s) = \hat{f}_{T_{\rm C}}(s+k_{-1})/\left(\frac{k_1[\mathbf{S}]+s}{k_1[\mathbf{S}]} - \frac{k_{-1}}{k_{-1}+s}[1 - \hat{f}_{T_{\rm C}}(s+k_{-1})]\right) (32)$$

The semi-Markov approximation thus provides a quick way to obtain the waiting time distribution f(t).

The Laplace inverse of eq 32 is not known in closed form for general $f_{T_{\rm C}}(t)$, but given an expression for $f_{T_{\rm C}}(t)$, f(t) can be readily calculated from eq 32 numerically. As an example, if the distribution $f_{T_{\rm C}}(t)$ were described by a sum of three exponentials:

$$f_{T_{\rm C}}(t) = \sum_{i=1}^{3} a_i \kappa_i \exp(-\kappa_i t)$$
(33)

with the a_i satisfying $\sum_{i=1}^{3} a_i = 1$ to ensure normalization of $f_{T_C}(t)$, the calculated f(t) as a function of [S], for some suitable set of parameters k_1 , k_{-1} , a_i , and κ_i , is easily shown to reproduce the general trends depicted in Figure 4. In other words, f(t) is a stretched multiexponential decay at high substrate concentration and a single-exponential decay at low concentrations, again in qualitative agreement with experiment.¹⁶

The average reaction rate $1/\langle t \rangle$ calculated from eq 32 is

$$\frac{1}{\langle t \rangle} = \frac{\gamma_2[S]}{[S] + \Delta_{\rm M}} \tag{34}$$

where

$$\gamma_2 = \frac{k_{-1} f_{T_{\rm C}}(k_{-1})}{1 - \hat{f}_{T_{\rm C}}(k_{-1})} \qquad \Delta_{\rm M} = \frac{k_{-1}}{k_1 [1 - \hat{f}_{T_{\rm C}}(k_{-1})]} \quad (35)$$

Here $\hat{f}_{T_{\rm C}}(k_{-1})$ means $\hat{f}_{T_{\rm C}}(s + k_{-1})|_{s=0}$. Parallel with the definition of $K_{\rm M} = (k_{-1} + k_2)/k_1$, it is readily shown that



Figure 6. Randomness parameter *r* vs [S] under the following conditions: (i) no dynamic disorder (full line), calculated using eq 38 with $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, $k_2 = 250 \text{ s}^{-1}$, and $k_{-1} = 50 \text{ s}^{-1}$; (ii) dynamic disorder present in the catalytic step (dotted line), calculated using eq 31, with $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, $k_{-1} = 50 \text{ s}^{-1}$, and a = 6, $b = 35 \text{ in } w(k_2)$, with the mean of k_2 assigned the value 175 s⁻¹; (iii) dynamic disorder present in the dissociation step (dashed line), calculated using eq 31 according to the method described in the text, with $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, $k_2 = 175 \text{ s}^{-1}$, and a = 6, b = 10 in $w(k_2)$, with the mean of k_{-1} assigned the value 50 s⁻¹.

As in the microscopic model of dynamic disorder given in section 3, the semi-Markovian approximation also leads, in this case directly, without the imposition of additional constraints, to the Michaelis-Menten equation, with the ensemble parameters k_2 and K_M replaced by quantities (γ_2 and Δ_M , respectively) related to the waiting time distribution of the catalytic step. The classic Michaelis-Menten parameters are recovered when $f_{T_C}(t)$ is described by a single exponential, $k_2 \exp(-k_2 t)$.

It is worth noting again that since the above treatment invokes the semi-Markov approximation, successive enzyme turnover times are uncorrelated; i.e., they exhibit no memory effects.

5. Randomness Parameter

The probability density f(t) characterizes the kinetics of singlemolecule enzymatic reactions completely, with the *n*th moment in general given by $\langle t^n \rangle \equiv \int_0^\infty dt f(t)t^n$. While the first moment of f(t) can be described by the single-molecule Michaelis– Menten equation, higher moments of f(t) contain more information.²³ Often it is convenient to evaluate the second moment of f(t), which is related to a randomness parameter *r*, defined as^{36,37}

$$=\frac{\langle t^2 \rangle - \langle t \rangle^2}{\langle t \rangle^2} \tag{37}$$

For a one-step Poisson process, $f(t) = k \exp(-kt)$, $\langle t \rangle = 1/k$, $\langle t^2 \rangle - \langle t \rangle^2 = 1/k^2$, therefore, r = 1. For multistep processes, assuming an identical rate constant *k* in *n* sequential rate-limiting steps, the variance of *t* in the numerator of eq 37 is n/k^2 , while in the denominator variance is n^2/k^2 . Hence r = 1/n. The greater the number of rate-limiting steps, the smaller is the value of *r*.

r

In eq 1, if the reaction steps are all characterized by exponentially distributed waiting time distributions (implying no dynamic disorder), r has been shown to be given by³⁶

$$r = \frac{(k_1[S] + k_2 + k_{-1})^2 - 2k_1k_2[S]}{(k_1[S] + k_2 + k_{-1})^2}$$
(38)

which is drawn as the full line in Figure 6. This curve may be interpreted as follows. At low substrate concentrations [S], r is unity because substrate binding is the rate-limiting step. As [S] increases, r decreases, reflecting the formation of the enzyme—substrate complex as an intermediate. At still higher [S], r returns to unity when the catalytic step becomes rate-limiting.

The [S] dependence of r can be quite different when dynamic disorder is present. Assuming that dynamic disorder is manifested in the catalytic step of the reaction through the distribution $w(k_2)$ governing the range of possible values of k_2 , the evaluation of r is carried out by using eq 31 to calculate the first and second moments of f(t). These moments are found to be

$$\langle t^{2} \rangle = \frac{2}{k_{1}^{2} [S]^{2}} + \frac{2}{(a-1)bk_{1}[S]} \left(1 + \frac{2k_{-1}}{k_{1}[S]} \right) + \frac{2}{(a-1)(a-2)b^{2}} \left(1 + \frac{k_{-1}}{k_{1}[S]} \right)^{2}$$
(39a)

$$\langle t \rangle^{2} = \frac{1}{k_{1}^{2}[S]^{2}} + \frac{2}{(a-1)bk_{1}[S]} \left(1 + \frac{k_{-1}}{k_{1}[S]} \right) + \frac{1}{(a-1)^{2}b^{2}} \left(1 + \frac{k_{-1}}{k_{1}[S]} \right)^{2}$$
(39b)

The variation of *r* with [S] as determined by the above expressions is shown in Figure 6 (dotted line) for the following parameter values: a = 6, b = 35, $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, and $k_{-1} = 50 \text{ s}^{-1}$, with the mean of k_2 [which is given by (a - 1)b] being 175 s⁻¹. As is evident, *r* can exceed 1 at high substrate concentration. This is in agreement with recent experimental findings.¹⁶ We should note that another reason for *r* being larger than unity is the existence of a reversible reaction in the catalytic step, which was previously reported,⁴² but can only occur for enzymatic reactions close to equilibrium.

It is conceivable that dynamic disorder could be manifested in the dissociation step of the reaction, in which case the rate constant k_{-1} , rather than k_2 , would have a range of different values, governed by a distribution function $w(k_{-1})$. The evaluation of r under these circumstances can be carried out, as before, by using eq 31 with $w(k_{-1})$ replacing $w(k_2)$, and the integration being performed over k_{-1} . If $w(k_{-1})$ is given by a gamma distribution, with a = 6, b = 10, (such that the mean of k_{-1} is 50 s⁻¹) and the other parameters are assigned the values $k_1 = 10^7 \text{ M}^{-1} \text{ s}^{-1}$, and $k_2 = 175 \text{ s}^{-1}$, the result of such a calculation is shown as the dashed line in Figure 6. Thus, when dynamic disorder is present in the dissociation step of the reaction only, r cannot exceed unity at high substrate concentrations. This discussion highlights the fact that if r is observed experimentally to be greater than 1 for an irreversible enzymatic reaction, then (i) dynamic disorder must be present, and (ii) it must be present in the catalytic step, because a constant k_2 cannot give rise to r larger than unity.

6. Conclusions

The classic Michaelis—Menten equation relates the enzymatic velocity to the substrate concentration. It was derived by solving the nonlinear differential eqs 2-4 under the steady-state approximation. Single-molecule enzymatic turnover experiments provide more dynamic information than the ensemble-averaged results through a different observable, f(t), the probability density of the waiting time.

In the absence of dynamic disorder, a single-molecule Michaelis-Menten equation, eq 10, that explicitly describes the

temporal behavior of f(t) at any specified substrate concentration is easily derived. The f(t) in eq 10 is exact and does not invoke the steady-state assumption, but it can be reduced to the steadystate case, i.e., to a fast rise followed by an exponential decay, when $k_2 \ll k_{-1}$. Irrespective of the use of the steady-state condition, the reciprocal of the first moment of f(t) is always consistent with the ensemble Michaelis–Menten equation (eq 5), which is an important insight.

In the presence of dynamic disorder, the treatment of singlemolecule Michaelis-Menten kinetics becomes considerably more complex. The existence of distinct conformational states of the enzyme that interconvert on time scales comparable to or longer than the time scales of the reaction results in disperse kinetics. The Michaelis-Menten mechanism of eq 1 is easily generalized to include these conformational states (eqs 15 and 28). The calculation of the waiting time distribution f(t) and the mean enzymatic rate $1/\langle t \rangle$ for these multistate models of dynamic disorder can be carried out analytically. It has been found that $1/\langle t \rangle$ does not always exhibit the same substrate concentration dependence as the Michaelis-Menten equation. However, under many conditions, $1/\langle t \rangle$ does follow the single-molecule Michaelis-Menten equation. In these limits, the parameters k_2 and K_M that appear in the ensemble Michaelis-Menten equation are replaced by the weighted averages of distributions of the corresponding kinetic constants of the conformers. Although the first moment of f(t) contains the same information as in the ensemble measurements, f(t) itself provides new informantion about dynamic disorder, and it also exhibits multiexponential long tails under saturating substrate concentrations.

A semi-Markovian approximation to this description of singlemolecule kinetics views the origin of dynamic disorder in terms of the occurrence of a non-Poisson distribution of reaction times in the catalytic step of eq 1. In this approximation, the calculated mean rate, $1/\langle t \rangle$, directly recovers the hyperbolic substrate concentration dependence of the ensemble Michaelis-Menten equation. As before, the parameters k_2 and K_M are replaced by generalized counterparts. At the same time, the waiting time distribution f(t) is now found to show highly nonexponential decay at high substrate concentrations, as seen in experiments.

The first and second moments of f(t) (when calculated with the model described by eq 31) can be used to find expressions for the randomness parameter, r, which provides a convenient characterization of single-molecule turnover trajectories. In the absence of dynamic disorder, if the reaction is dominated by one rate-limiting step, r = 1, whereas if the reaction has more than one rate-limiting step, r < 1. In the presence of dynamic disorder, however, r could be greater than 1.

We hope the results in this paper provide a theoretical framework for understanding the ever-expanding activities in single-molecule enzymology, and perhaps enzymology in general.

Acknowledgment. We are grateful to Attila Szabo and Hong Qian for their critical reading of the manuscript and stimulating comments. We also thank Liang Jiang for helpful comments, and Joseph Klafter for sending us preprints of refs 14 and 15. X.S.X. acknowledges NIH Roadmap for Medical Research/NIH Director's Pioneer Award Program for support, and S.C.K. acknowledges support from NSF grant DMS-02-04674 and the NSF Career Award.

Note Added in Proof. We give in Section 3(i) six limiting conditions (a-f) under which the single-molecule Michaelis-Menten equation holds for the two-state model (eq 15). To be more precise, in condition (a) we mean $\beta/k_{21} \rightarrow 0$ and $\beta/k_{22} \rightarrow 0$, but α/k_{21} and α/k_{22} do not approach 0; in condition (b) we mean $\beta \rightarrow 0$, but α does not approach 0; in condition (c) we mean $\alpha \rightarrow 0$, but β does not approach 0. Correspondingly, the six limiting conditions (a-f) in Section 3(ii) for the single-molecule Michaelis-Menten equation to hold for the multistate model (eq 28) should be more precisely stated in a similar way.

Appendix A. Calculation of the Waiting Time Distribution for the Two-State Model

The total time to form the product P in the two-state kinetic scheme of eq 15 is a random variable T that is governed by the waiting time distribution f(t). To calculate f(t), the quantity of interest, we first seek expressions for the waiting time distributions that govern the times needed to complete the reaction starting from E₁, E₂, ES₁, or ES₂. These times, which are random variables, are denoted T_{E_1} , T_{E_2} , T_{ES_1} , and T_{ES_2} , respectively, and their corresponding waiting time distributions are denoted $f_{T_{E_1}}(t)$, $f_{T_{ES_1}}(t)$, and $f_{T_{ES_2}}(t)$. Once these distributions are determined, f(t) can be expressed as a weighted steady-state average of $f_{T_{E_1}}(t)$ and $f_{T_{ES_2}}(t)$, we set up and solve a set of four linear simultaneous equations as follows.

Imagine that the system is initially in the state E_1 , so that the total time to complete the reaction and form the product is the random variable T_{E_1} . From E_1 the system can proceed either to E_2 or to ES_1 , from where the reaction is then completed in the time T_{E_2} or T_{ES_1} . The system reaches E_2 if the step $E_1 \rightarrow E_2$ occurs before the step $E_1 \rightarrow ES_1$ (or equivalently, if the time T_{α} to complete the former is less than the time T_{11} to complete the latter), and it reaches ES_1 if the step $E_1 \rightarrow ES_1$ occurs before the step $E_1 \rightarrow E_2$ (or equivalently, if $T_{11} < T_{\alpha}$.) Hence, the probability that T_{E_1} is realized within some time interval *t*, which we denote $P(T_{E_1} < t)$, can be written as

$$P(T_{E_1} < t) = P(T_{E_2} + T_{\alpha} < t)P(T_{\alpha} < T_{11}) + P(T_{ES_1} + T_{11} < t)P(T_{11} < T_{\alpha})$$
(A.1)

The steps $E_1 \rightarrow E_2$ and $E_1 \rightarrow ES_1$ occur at random through a Poisson process with average rates α and $k_{11}[S]$, respectively. Hence, the times T_{α} and T_{11} are drawn from the following waiting time distributions:

$$f_{T_{\alpha}}(t) = \alpha \exp(-\alpha t)$$
 (A.2)

$$f_{T_{11}}(t) = k_{11}[S] \exp(-k_{11}[S]t)$$
 (A.3)

In general, for any random variable X, $f_X(x) = dP(X < x)/dx$, so eq A.1 can be differentiated with respect to t to produce

$$f_{T_{E_1}}(t) = f_{T_{E_2} + T_\alpha}(t) P(T_\alpha < T_{11}) + f_{T_{ES_1} + T_{11}}(t) P(T_{11} < T_\alpha)$$
(A.4)

Since the distribution of the sum of two random variables is the convolution of the distributions of the individual random variables, eq A.4 can be further written as

$$f_{T_{E_1}}(t) = \int_0^t dt_1 f_{T_{E_2}}(t-t_1) f_{T_\alpha}(t_1) P(t_1 < T_{11}) + \int_0^t dt_1 f_{T_{ES_1}}(t-t_1) f_{T_{11}}(t_1) P(t_1 < T_\alpha)$$
(A.5)

From eqs A.2 and A.3, we can show that

$$P(t_1 < T_{11}) = k_{11}[S] \int_{t_1}^{\infty} dt_2 \exp(-k_{11}[S]t_2) = \exp(-k_{11}[S]t_1) \quad (A.6)$$

$$P(t_1 < T_{\alpha}) = \alpha \int_{t_1}^{\infty} dt_2 \exp(-\alpha t_2) = \exp(-\alpha t_1) \quad (A.7)$$

Therefore, by taking the Laplace transform of eq A.5, and making use of eqs A.2 and A.3 and A.6 and A.7, we obtain

$$\hat{f}_{T_{E_1}}(s) = \frac{\alpha \hat{f}_{T_{E_2}}(s)}{s + \alpha + k_{11}[S]} + \frac{k_{11}[S]\hat{f}_{T_{ES_1}}(s)}{s + \alpha + k_{11}[S]}$$
(A.8)

where $\hat{f}_{T_{E_1}}(s)$, $\hat{f}_{T_{E_2}}(s)$, and $\hat{f}_{T_{ES_1}}(s)$ are Laplace transforms of $f_{T_{E_1}}(t)$, $f_{T_{E_2}}(t)$, and $f_{T_{ES_1}}(t)$, respectively. Equation A.8 is one equation connecting three of the unknown waiting time distributions, $f_{T_{E_1}}(t)$, $f_{T_{E_2}}(t)$ and $f_{T_{ES_1}}(t)$. An equation for the waiting time distribution of the random variable T_{E_2} , the time to complete the reaction starting from the state E_2 , can be obtained immediately from eq A.8 by an interchange of labels. That is,

$$\hat{f}_{T_{E_2}}(s) = \frac{\alpha \hat{f}_{T_{E_1}}(s)}{s + \alpha + k_{12}[S]} + \frac{k_{12}[S] \hat{f}_{T_{ES_2}}(s)}{s + \alpha + k_{12}[S]}$$
(A.9)

We can also derive two more expressions involving $\hat{f}_{T_{\text{ES}_1}}(s)$ and $\hat{f}_{T_{\text{ES}_2}}(s)$ in a similar way. In outline, the procedure is as follows. Imagine the system to start in the state ES₁, so that T_{ES_1} is the time needed to complete the reaction and form the product. At ES₁, the system can either dissociate and return to E₁ at a rate constant k_{-11} , or it can isomerize to the conformer ES₂ at a rate constant β , or it can catalyze the substrate to the product at a rate constant k_{21} . The probability that the reaction time T_{ES_1} occurs within a time interval *t* (following the earlier reasoning) is therefore given by

$$\begin{split} P(T_{\text{ES}_{1}} < t) = \\ P(T_{-11} + T_{\text{E}_{1}} < t) P(T_{-11} < T_{\beta}) P(T_{-11} < T_{21}) + \\ P(T_{\beta} + T_{\text{ES}_{2}} < t) P(T_{\beta} < T_{-11}) P(T_{\beta} < T_{21}) + \\ P(T_{21} < t) P(T_{21} < T_{-11}) P(T_{21} < T_{\beta}) \quad (A.10) \end{split}$$

where T_{-11} , T_{β} , and T_{21} are the random times required to execute the steps $\text{ES}_1 \rightarrow \text{E}_1$, $\text{ES}_1 \rightarrow \text{ES}_2$, and $\text{ES}_1 \rightarrow \text{E}_1^0$, respectively. As before, these steps are Poisson processes, so T_{-11} , T_{β} , and T_{21} are drawn from the waiting time distributions

$$f_{T_{-1}}(t) = k_{-11} \exp(-k_{-11}t) \tag{A.11}$$

$$f_{T_a}(t) = \beta \exp(-\beta t) \tag{A.12}$$

$$f_{T_{21}}(t) = k_{21} \exp(-k_{21}t) \tag{A.13}$$

After differentiating eq A.10 with respect to t, making use of eqs A.11–A.13 and taking Laplace transforms, it is readily shown that

$$\hat{f}_{T_{\text{ES}_{1}}}(s) = \frac{k_{-11}\hat{f}_{T_{\text{E}_{1}}}(s)}{s+\beta+k_{-11}+k_{21}} + \frac{\beta\hat{f}_{T_{\text{ES}_{2}}}(s)}{s+\beta+k_{-11}+k_{21}} + \frac{k_{21}}{\frac{k_{21}}{s+\beta+k_{-11}+k_{21}}}$$
(A.14)

This is a third equation connecting the unknown waiting time

distributions. A fourth and final equation is obtained from eq A.14 by symmetry:

$$\hat{f}_{T_{\text{ES}_{2}}}(s) = \frac{k_{-12}\hat{f}_{T_{\text{E}_{2}}}(s)}{s+\beta+k_{-12}+k_{22}} + \frac{\beta\hat{f}_{T_{\text{ES}_{1}}}(s)}{s+\beta+k_{-12}+k_{22}} + \frac{k_{22}}{\frac{k_{22}}{s+\beta+k_{-12}+k_{22}}}$$
(A.15)

Equations A.8, A.9, A.14, and A.15 are conveniently represented in matrix form as

$$s\hat{\mathbf{f}}(s) = \mathbf{Q}\hat{\mathbf{f}}(s) + \mathbf{r}$$
 (A.16)

where $\hat{\mathbf{f}}(s) = (\hat{f}_{T_{E_1}}(s), \hat{f}_{T_{E_2}}(s), \hat{f}_{T_{ES_1}}(s), \hat{f}_{T_{ES_2}}(s))^{\mathrm{T}}, \mathbf{r} = (0, 0, k_{21}, k_{22})^{\mathrm{T}}$, and

 $\mathbf{Q} =$

$$\begin{pmatrix} -(\alpha + k_{11}[S]) & \alpha & k_{11}[S] & 0 \\ \alpha & -(\alpha + k_{12}[S]) & 0 & k_{12}[S] \\ k_{-11} & 0 & -(\beta + k_{-11} + k_{21}) & \beta \\ 0 & k_{-12} & \beta & -(\beta + k_{-12} + k_{22}) \\ \end{pmatrix}$$
(A.17)

Equation A.16 can be inverted to yield the following expression for $\hat{\mathbf{f}}(s)$ [cf. eq 18]:

$$\hat{\mathbf{f}}(s) = (s\mathbf{I} - \mathbf{Q})^{-1}\mathbf{r} \tag{A.18}$$

The relation between $\hat{\mathbf{f}}(s)$ and f(t), the waiting time distribution that is actually measured in experiments, is now established as follows. At the end of each catalytic cycle, the enzyme exists either as the conformer E_1^0 or as the conformer E_2^0 . E_1^0 is assumed to return to E_1 at a rate constant δ_1 , while E_2^0 is assumed to return to E_2 at a rate constant δ_2 . In single-molecule enzymatic turnover experiments, the successive reaction times are obtained over a long time interval so that many turnovers occur. But over the course of many such turnover cycles, the fraction of time that the enzyme resides in E_1 or E_2 attains a steady-state value. Therefore, during a long time trajectory, the waiting time distribution f(t) observed in enzymatic turnover experiments corresponds to the steady-state weighted average of the two waiting time distributions $f_{T_{E_1}}(t)$ and $f_{T_{E_2}}(t)$, when the reaction starts from E_1 or E_2 , respectively. This steady-state weight for $f_{T_{r_i}}(t)$, which accounts for how often the system starts from E1 immediately after restarting the cycle, is proportional to the steady-state probability $P_{E_1^0}^0$ that the system is in E_1^0 multiplied by the rate constant δ_1 of E_1^{0} 's return to the E_1 state. Similarly, the steady-state weight for $f_{T_{E_2}}(t)$ is proportional to the steady-state probability $P_{E_2^0}^0$ that the system is in E_2^0 multiplied by the rate constant δ_2 of E_2^0 's return to the E_2 state. In other words

$$f(t) = \frac{\delta_1 P_{E_1^0}^0}{\delta_1 P_{E_1^0}^0 + \delta_2 P_{E_2^0}^0} f_{T_{E_1}}(t) + \frac{\delta_2 P_{E_2^0}^0}{\delta_1 P_{E_1^0}^0 + \delta_2 P_{E_2^0}^0} f_{T_{E_2}}(t)$$
(A.19)

where the denominator in this expression is introduced to ensure that f(t) is properly normalized to unity.

Equation A.19 can be rewritten in Laplace space in terms of the vector $\hat{\mathbf{f}}(s)$ obtained earlier:

$$\hat{f}(s) = \left(\frac{\delta_1 P^0_{E_1^0}}{\delta_1 P^0_{E_1^0} + \delta_2 P^0_{E_2^0}}, \frac{\delta_2 P^0_{E_2^0}}{\delta_1 P^0_{E_1^0} + \delta_2 P^0_{E_2^0}}, 0, 0\right) \hat{\mathbf{f}}(s)$$
(A.20)

The calculation of the equilibrium probabilities $P_{E_1^0}^0$ and $P_{E_2^0}^0$ is discussed next.

In the kinetic scheme shown in eq 15, the probabilities $P_{\nu}(t)$ that the enzyme is in one or other of the states $\nu = E_1, E_2, ES_1, ES_2, E_1^{0}$, or E_2^{0} satisfy a master equation, i.e., a system of coupled linear first order differential equations analogous to eqs 6–8 that express the balance of probability into and out of ν . At long times, $t \rightarrow \infty$, when the system reaches equilibrium, the rates $dP_{\nu}(t)/dt$ vanish, and the above system of equations reduces to

$$\mathbf{SP} = 0 \tag{A.21}$$

where **P** is the vector $(P_{E_1}^0, P_{E_2}^0, P_{ES_1}^0, P_{ES_2}^0, P_{E_1^{00}}^0, P_{E_2^{00}}^0)^T$, the superscript 0 denoting the equilibrium value, and

S	=						
	$-(\alpha + k_{11}[S])$	α	k_{-11}	0	δ_1	0	
	α	$-(\alpha + k_{12}[\mathbf{S}])$	0	k_{-12}	0	δ_2	
	$k_{11}[S]$	0	$-(\beta + k_{-11} + k_{21})$	β	0	0	l
	0	$k_{12}[S]$	β	$-(\beta + k_{-12} + k_{22})$	0	0	I
	0	0	k_{21}	0	$-(\gamma + \delta_1)$	γ	l
	0	0	0	k ₂₂	γ	$-(\gamma + \delta_2)$	
						(A.22)

From the solution to eq A.21 under the constraint

$$P_{\rm E_1}^0 + P_{\rm E_2}^0 + P_{\rm ES_1}^0 + P_{\rm ES_2}^0 + P_{\rm E_1^0}^0 + P_{\rm E_2^0}^0 = 1 \quad (A.23)$$

and in the limit δ_1 , $\delta_2 \gg 1$ (corresponding to fast reset of E_1^0 and E_2^0 to E_1 and E_2), it can be shown that

$$\frac{\delta_{1}P_{E_{1}^{0}}^{0}}{\delta_{2}P_{E_{2}^{0}}^{0}} = \frac{k_{21}[\alpha(k_{11}k_{22} + k_{11}k_{-12}) + \alpha\beta(k_{11} + k_{12}) + \beta k_{11}k_{12}[S]]}{k_{22}[\alpha(k_{12}k_{21} + k_{12}k_{-11}) + \alpha\beta(k_{11} + k_{12}) + \beta k_{11}k_{12}[S]]}$$
(A.24)

from which the steady-state fractions in eq A.20 can be calculated. Once we obtain the Laplace transform $\hat{f}(s)$, the mean waiting time $\langle t \rangle = \int_0^\infty dt \ tf(t)$ is readily given by $\langle t \rangle = -d/ds \ \hat{f}(s)|_{s=0}$. Using eqs A.18 and A.20, we have

$$\langle t \rangle = \left(\frac{\delta_1 P_{E_1^0}^0}{\delta_1 P_{E_1^0}^0 + \delta_2 P_{E_2^0}^0}, \frac{\delta_2 P_{E_2^0}^0}{\delta_1 P_{E_1^0}^0 + \delta_2 P_{E_2^0}^0}, 0, 0 \right) \mathbf{Q}^{-2} \mathbf{r}$$
(A.25)

which, after lengthy but straightforward algebra, can be rearranged to the form in eq 24.

Appendix B. Calculation of the Waiting Time Distribution for the Multistate Case

The approach introduced in Appendix A can also be applied to the analysis of the multistate model of dynamic disorder. In deriving an expression for the waiting time distribution f(t) and its first moment, it is convenient to introduce some simplified notation. Let A_i , B_i , and C_i stand for the states E_i , ES_i , and E_i^0 , respectively. As diagrammed in eq 28, A_i can convert to B_i or to any of the A_j , B_i can convert to A_i , C_i , or to any of the B_j , and C_i can convert to A_i or to any of the C_j . If we introduce the matrix \mathbf{Q}_{AA} to denote the transition rates between the A_i 's and the matrix \mathbf{Q}_{AB} to denote the transition rates between the A_i 's and the B_i 's, and likewise introduce \mathbf{Q}_{BA} , \mathbf{Q}_{BB} , \mathbf{Q}_{CC} , and \mathbf{Q}_{CA} to denote the transition rates between the corresponding states, then, following the diagram of eq 28, the transition matrices are $\mathbf{Q}_{AA} = [\alpha_{ij}]$, $\mathbf{Q}_{BB} = [\beta_{ij}]$, $\mathbf{Q}_{CC} = [\gamma_{ij}]$, $\mathbf{Q}_{AB} = \text{diag}(k_{11}[S]$, $k_{12}[S]$, ..., $k_{1n}[S]$), $\mathbf{Q}_{BA} = \text{diag}(k_{-11}, k_{-12}, ..., k_{-1n})$, $\mathbf{Q}_{BC} =$ diag $(k_{21}, k_{22}, ..., k_{2n})$, and $\mathbf{Q}_{CA} = \text{diag}(\delta_1, \delta_2, ..., \delta_n)$. The entire network of interconversions can be described by a matrix \mathbf{Q} given by

$$\mathbf{Q} = \begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} & \mathbf{Q}_{AB} & \mathbf{0} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}) & \mathbf{Q}_{BC} \\ \mathbf{Q}_{CA} & \mathbf{0} & \mathbf{Q}_{CC} - \mathbf{Q}_{CA} \end{pmatrix}$$
(B.1)

Following the method of Appendix A, one can derive the following relation, in Laplace space, between the distributions associated with the times needed to complete the reaction starting from different states of the network:

$$\begin{pmatrix} \hat{\mathbf{f}}_{T_{A}}(s) \\ \hat{\mathbf{f}}_{T_{B}}(s) \end{pmatrix} = \begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} & \mathbf{Q}_{AB} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}) \end{pmatrix} \begin{pmatrix} \hat{\mathbf{f}}_{T_{A}}(s) \\ \hat{\mathbf{f}}_{T_{B}}(s) \end{pmatrix} + \begin{pmatrix} \mathbf{0} \\ \mathbf{Q}_{BC} \mathbf{1} \end{pmatrix} (B.2)$$

where $\hat{\mathbf{f}}_{T_{A}}(s) = (\hat{f}_{T_{A_{1}}}(s), \hat{f}_{T_{A_{2}}}(s), ..., \hat{f}_{T_{A_{n}}}(s))^{\mathrm{T}}$, $\hat{\mathbf{f}}_{T_{\mathrm{B}}}(s) = (\hat{f}_{T_{\mathrm{B}_{1}}}(s), \hat{f}_{T_{\mathrm{B}_{2}}}(s), ..., \hat{f}_{T_{\mathrm{B}_{n}}}(s))^{\mathrm{T}}$, and $\mathbf{1} = (1, 1, ..., 1)^{\mathrm{T}}$. Here $\hat{f}_{T_{A_{i}}}(s)$ and $\hat{f}_{T_{\mathrm{B}_{i}}}(s)$ are the Laplace transforms of the waiting time distributions where the system starts from A_{i} and B_{i} , respectively. Equation B.2 can be solved by matrix inversion:

To calculate the mean waiting time $\langle t \rangle$, eq B.2 is differentiated with respect to *s*, and the result evaluated at s = 0. Together with eq B.3, this leads to

$$-\begin{pmatrix} \hat{\mathbf{f}}'_{T_{A}}(0) \\ \hat{\mathbf{f}}'_{T_{B}}(0) \end{pmatrix} = -\begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} & \mathbf{Q}_{AB} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}) \end{pmatrix}^{-1} \begin{pmatrix} \hat{\mathbf{f}}_{T_{A}}(0) \\ \hat{\mathbf{f}}_{T_{B}}(0) \end{pmatrix}$$
$$= -\begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} & \mathbf{Q}_{AB} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}) \end{pmatrix}^{-1} \begin{pmatrix} \mathbf{1} \\ \mathbf{1} \end{pmatrix}$$
(B.4)

where the last equality makes use of the results $\hat{f}_{T_{A_i}}(0) = \hat{f}_{T_{B_i}}(0)$ = 1. The inverse in eq B.4 is calculated by block matrix inversion, producing

$$\begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} & \mathbf{Q}_{AB} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}) \end{pmatrix}^{-1} = \begin{pmatrix} \mathbf{L} & \mathbf{M} \\ \mathbf{N} & \mathbf{R} \end{pmatrix} \quad (B.5)$$

where

$$\mathbf{L} = [\mathbf{Q}_{AA} - \mathbf{Q}_{AB} - \mathbf{Q}_{AB}(\mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}))^{-1}\mathbf{Q}_{BA}]^{-1}$$
(B.6)

$$\mathbf{M} = -[\mathbf{Q}_{AA} - \mathbf{Q}_{AB} - \mathbf{Q}_{AB}(\mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}))^{-1}\mathbf{Q}_{BA}]^{-1}\mathbf{Q}_{AB}(\mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}))^{-1} (B.7)$$

$$\mathbf{N} = -(\mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}))^{-1}\mathbf{Q}_{BA}[\mathbf{Q}_{AA} - \mathbf{Q}_{AB} - \mathbf{Q}_{AB}]^{-1}\mathbf{Q}_{AB}(\mathbf{Q}_{BB} - (\mathbf{Q}_{BA} + \mathbf{Q}_{BC}))^{-1}\mathbf{Q}_{BA}]^{-1}$$
(B.8)

$$\mathbf{R} = [\mathbf{Q}_{BB} - \mathbf{Q}_{BA} - \mathbf{Q}_{BC} - \mathbf{Q}_{BA}(\mathbf{Q}_{AA} - \mathbf{Q}_{AB})^{-1}\mathbf{Q}_{AB}]^{-1}$$
(B.9)

It then follows that

$$\begin{pmatrix} \hat{\mathbf{f}}'_{T_{A}}(0) \\ \hat{\mathbf{f}}'_{T_{B}}(0) \end{pmatrix} = \begin{pmatrix} (\mathbf{L} + \mathbf{M})\mathbf{1} \\ (\mathbf{N} + \mathbf{R})\mathbf{1} \end{pmatrix}$$
(B.10)

As before, since each new reaction cycle can start from any A_i , the overall waiting time distribution f(t) is the steady-state weighted average of the distributions $f_{T_{A_i}}(t)$. The calculation of these steady-state probabilities is considered next. Let P_{A_i} , P_{B_i} , and P_{C_i} denote the steady-state probabilities of A_i , B_i , and C_i , respectively. To calculate them, we proceed as earlier from the stationary solution of the master equation, which is defined by

$$(\mathbf{p}_{\mathrm{A}}^{\mathrm{T}} \quad \mathbf{p}_{\mathrm{B}}^{\mathrm{T}} \quad \mathbf{p}_{\mathrm{C}}^{\mathrm{T}})\mathbf{Q} = 0 \tag{B.11}$$

where $\mathbf{p}_{A} = (P_{A_{1}}, P_{A_{2}}, ..., P_{A_{n}})^{T}$, $\mathbf{p}_{B} = (P_{B_{1}}, P_{B_{2}}, ..., P_{B_{n}})^{T}$, $\mathbf{p}_{C} = (P_{C_{1}}, P_{C_{2}}, ..., P_{C_{n}})^{T}$. To solve eq B.11, \mathbf{p}_{A} and \mathbf{p}_{B} are first rewritten in terms of \mathbf{p}_{C} , so that an equation solely in \mathbf{p}_{C} is obtained. This equation is then solved by standard matrix techniques. To implement the first step in this process, the definition of \mathbf{Q} is substituted into eq B.11, producing

$$(\mathbf{p}_{A}^{T} \ \mathbf{p}_{B}^{T} \ \mathbf{p}_{C}^{T}) \begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} \ \mathbf{Q}_{AB} & \mathbf{Q}_{AB} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - \mathbf{Q}_{BA} - \mathbf{Q}_{BC} & \mathbf{Q}_{BC} \\ \mathbf{Q}_{CA} & \mathbf{0} & \mathbf{Q}_{CC} - \mathbf{Q}_{CA} \end{pmatrix} = \mathbf{0}$$

$$(B.12)$$

Equation B.12 implies that

$$(\mathbf{p}_{A}^{T} \ \mathbf{p}_{B}^{T}) \begin{pmatrix} \mathbf{Q}_{AA} - \mathbf{Q}_{AB} & \mathbf{Q}_{AB} \\ \mathbf{Q}_{BA} & \mathbf{Q}_{BB} - \mathbf{Q}_{BA} - \mathbf{Q}_{BC} \end{pmatrix} + \mathbf{p}_{C}^{T} (\mathbf{Q}_{CA} \ \mathbf{0}) = \mathbf{0}$$
(B.13)

and

$$\mathbf{p}_{\rm B}^{\rm T} \mathbf{Q}_{\rm BC} + \mathbf{p}_{\rm C}^{\rm T} (\mathbf{Q}_{\rm CC} - \mathbf{Q}_{\rm CA}) = \mathbf{0}$$
(B.14)

Equation B.13 leads to

Hence,

$$\mathbf{p}_{A}^{T} = -\mathbf{p}_{C}^{T}\mathbf{Q}_{CA}\mathbf{L}$$
 $\mathbf{p}_{B}^{T} = -\mathbf{p}_{C}^{T}\mathbf{Q}_{CA}\mathbf{M}$ (B.16)

Substituting eq B.16 into eq B.14, the sought for equation in $\mathbf{p}_{\rm C}$ is obtained as

$$-\mathbf{p}_{\mathrm{C}}^{\mathrm{T}}\mathbf{Q}_{\mathrm{CA}}\mathbf{M}\mathbf{Q}_{\mathrm{BC}} + \mathbf{p}_{\mathrm{C}}^{\mathrm{T}}(\mathbf{Q}_{\mathrm{CC}} - \mathbf{Q}_{\mathrm{CA}}) = \mathbf{0} \qquad (B.17)$$

which can be solved for \mathbf{p}_{C} up to a normalizing constant. This then provides expressions for \mathbf{p}_{A} and \mathbf{p}_{B} via eq B.16.

Having obtained the steady-state probabilities, the overall waiting time distribution f(t) (in Laplace space) is calculated from

$$\hat{f}(s) = \frac{\mathbf{p}_{\mathrm{C}}^{\mathrm{T}} \mathbf{Q}_{\mathrm{CA}} \hat{\mathbf{f}}_{T_{\mathrm{A}}}(s)}{\mathbf{p}_{\mathrm{C}}^{\mathrm{T}} \mathbf{Q}_{\mathrm{CA}} \mathbf{1}}$$
(B.18)

Hence, the mean waiting time is calculated as

$$\langle t \rangle = \frac{-\mathbf{p}_{C}^{T} \mathbf{Q}_{CA} \hat{\mathbf{f}}'_{T_{A}}(0)}{\mathbf{p}_{C}^{T} \mathbf{Q}_{CA} \mathbf{1}}$$
$$= \frac{-\mathbf{p}_{C}^{T} \mathbf{Q}_{CA} (\mathbf{L} + \mathbf{M}) \mathbf{1}}{\mathbf{p}_{C}^{T} \mathbf{Q}_{CA} \mathbf{1}}$$
(B.19)

Introducing a vector \mathbf{v}^{T} defined as $\mathbf{v}^{T} = \mathbf{p}_{C}^{T} \mathbf{Q}_{CA}$, eqs B.18 and B.19 can be written in more compact notation as

$$\hat{\mathbf{f}}(s) = \frac{\mathbf{v}^{\mathrm{T}} \mathbf{f}_{T_{\mathrm{A}}}(s)}{\mathbf{v}^{\mathrm{T}} \mathbf{1}} \qquad \langle t \rangle = \frac{-\mathbf{v}^{\mathrm{T}} (\mathbf{L} + \mathbf{M}) \mathbf{1}}{\mathbf{v}^{\mathrm{T}} \mathbf{1}} \quad (B.20)$$

Hence, the steady-state waiting time distribution f(t) is the weighted average of the waiting time distributions associated with starting the reaction from E_i , with the weights given by the steady-state probability to be in E_i .

Now from eqs B.6–B.9, it follows that

$$\mathbf{L} = -\mathbf{M}(\mathbf{Q}_{BB} - \mathbf{Q}_{BA} - \mathbf{Q}_{BC})\mathbf{Q}_{AB}^{-1} \qquad (B.21)$$

which leads to

$$\mathbf{L} + \mathbf{M} = -\mathbf{M}[(\mathbf{Q}_{BB} - \mathbf{Q}_{BA} - \mathbf{Q}_{BC})\mathbf{Q}_{AB}^{-1} - \mathbf{I}] \quad (B.22)$$

This equation, together with eq B.17, yields

$$\langle t \rangle = \frac{-\mathbf{v}^{\mathrm{T}}(\mathbf{L} + \mathbf{M})\mathbf{1}}{\mathbf{v}^{\mathrm{T}}\mathbf{1}}$$
$$= \frac{1}{\mathbf{v}^{\mathrm{T}}\mathbf{1}}\mathbf{v}^{\mathrm{T}}\mathbf{Q}_{\mathrm{CA}}^{-1}(\mathbf{Q}_{\mathrm{CC}} - \mathbf{Q}_{\mathrm{CA}})\mathbf{Q}_{\mathrm{BC}}^{-1}[(\mathbf{Q}_{\mathrm{BB}} - \mathbf{Q}_{\mathrm{BA}} - \mathbf{Q}_{\mathrm{BC}})\mathbf{Q}_{\mathrm{AB}}^{-1} - \mathbf{I}]\mathbf{1} \quad (B.23)$$

In the limit when the δ_i are much larger than the other rates (so that at the end of the reaction, the system returns rapidly to the state E_i), eq B.23 reduces to

$$\langle t \rangle = -\frac{1}{\mathbf{v}^{\mathrm{T}} \mathbf{1}} \mathbf{v}^{\mathrm{T}} \mathbf{Q}_{\mathrm{BC}}^{-1} [(\mathbf{Q}_{\mathrm{BB}} - \mathbf{Q}_{\mathrm{BA}} - \mathbf{Q}_{\mathrm{BC}}) \mathbf{Q}_{\mathrm{AB}}^{-1} - \mathbf{I}] \mathbf{1}$$
(B.24)

with **v** satisfying (cf. eq B.17) $\mathbf{v}^T \mathbf{M} \mathbf{Q}_{BC} + \mathbf{v}^T \mathbf{I} = \mathbf{0}$, or equivalently

$$\mathbf{v}^{\mathrm{T}}(\mathbf{I} + \mathbf{Q}_{\mathrm{BC}}^{-1}\mathbf{M}^{-1}) = \mathbf{0}$$
 (B.25)

Next we note that the mean enzymatic reaction rate, $1/\langle t \rangle$, calculated from eq B.24 will be of the Michaelis–Menten form (cf. eq 29)

$$\frac{1}{\langle t \rangle} = \frac{\chi_2[\mathbf{S}]}{[\mathbf{S}] + C_{\mathbf{M}}} \tag{B.26}$$

if the equilibrium weight **v** does not depend on the substrate concentration. This can be demonstrated by first noting that only the transition rates associated with $A_i \rightarrow B_i$ involve [S]. The matrix, \mathbf{Q}_{AB} can therefore be written $\mathbf{Q}_{AB} = [S]\mathbf{\tilde{Q}}_{AB}$, where $\mathbf{\tilde{Q}}_{AB} = \text{diag}(k_{11}, k_{12}, ..., k_{1n})$. The use of this definition in eq B.24 followed by rearrangement leads to eq B.26, with both χ_2 and C_M independent of [S]. It is now straightforward to show that each of the conditions (a-f) in section 3 (ii) does indeed guarantee that the weights **v** do not depend on the concentration [S], and that each condition therefore leads to a Michaelis–Menten equation, as seen in eq 29.

To calculate the waiting time distribution f(t) in the slow interconversion limit, we start from eq B.2, which can be written as

$$s\hat{\mathbf{f}}_{T_{A}}(s) = (\mathbf{Q}_{AA} - \mathbf{Q}_{AB})\hat{\mathbf{f}}_{T_{A}}(s) + \mathbf{Q}_{AB}\hat{\mathbf{f}}_{T_{B}}(s)$$
$$s\hat{\mathbf{f}}_{T_{B}}(s) = \mathbf{Q}_{BA}\hat{\mathbf{f}}_{T_{A}}(s) + (\mathbf{Q}_{BB} - \mathbf{Q}_{BA} - \mathbf{Q}_{BC})\hat{\mathbf{f}}_{T_{B}}(s) + \mathbf{Q}_{BC}\mathbf{1}$$
(B.27)

In the given limit, Q_{AA} and Q_{BB} are small, so eq B.27 effectively reduces to a set of equations for the individual components, i.e.,

$$\hat{sf}_{T_{A_i}}(s) = -k_{1i}[\mathbf{S}]\hat{f}_{T_{A_i}}(s) + k_{1i}[\mathbf{S}]\hat{f}_{T_{B_i}}(s)$$
$$\hat{sf}_{T_{B_i}}(s) = -k_{-1i}\hat{f}_{T_{A_i}}(s) - (k_{-1i} + k_{2i})\hat{f}_{T_{B_i}}(s) + k_{2i} \quad (B.28)$$

Solving for $\hat{f}_{T_{A_i}}(s)$, we obtain

$$\hat{f}_{T_{A_i}}(s) = \frac{k_{1i}k_{2i}[S]}{s^2 + s(k_{1i}[S] + k_{-1i} + k_{2i}) + k_{1i}k_{2i}[S]} \quad (B.29)$$

which can be inverted to

$$f_{T_{A_i}}(t) = \frac{k_{1i}k_{2i}[S]}{2\alpha_i} [\exp(A_i + B_i)t - \exp(B_i - A_i)t] \quad (B.30)$$

where A_i and B_i have been defined after eq 26. This expression, when weighted by w_i and summed over the reaction channels yields eq 30.

Appendix C. Calculation of the Waiting Time Distribution for the semi-Markovian Approximation

The general method of calculating f(t) remains the same as the method described in Appendix A. The reaction is imagined to start from **E**. The total time to complete the reaction starting from **E** is a random variable *T* governed by a waiting time distribution f(t). After a time T_1 , which is drawn from the waiting time distribution $f_{T_1}(t)$, the system moves to ES, from where the reaction is completed in a total time T_2 , which is drawn from the waiting time distribution $f_{T_2}(t)$. Thus, the probability that, starting from E, the reaction time *T* is realized within a time interval *t* is given by

$$P(T < t) = P(T_1 + T_2 < t)$$
(C.1)

The distribution $f_{T_1}(t)$ is assumed to be exponential, with a time constant of $k_1[S]$; hence $f_{T_1}(t) = k_1[S] \exp(-k_1[S]t)$. Differentiating eq C.1 with respect to *t*, we obtain

$$f(t) = k_1[\mathbf{S}] \int_0^t dt_1 \exp[-k_1[\mathbf{S}](t-t_1)] f_{T_2}(t_1) \quad (C.2)$$

The Laplace transform of eq C.2 yields

$$\hat{f}(s) = \frac{k_1[\mathbf{S}]}{s + k_1[\mathbf{S}]} \hat{f}_{T_2}(s)$$
 (C.3)

Once the enzyme has reached **ES**, a total time of T_2 will elapse before the product is formed. In forming the product, **ES** can either move to the state \mathbf{E}^0 directly in a time T_C drawn from the unknown waiting time distribution $f_{T_C}(t)$, or it can return to **E** in a time T_{-1} drawn from the exponential waiting time distribution $f_{T_{-1}}(t) = k_{-1} \exp(-k_{-1}t)$. The first option is selected if $T_C < T_{-1}$, the second if $T_{-1} < T_C$. The probability that, starting from **ES**, the reaction time T_2 is realized within a time *t* is therefore given by

$$P(T_2 < t) = P(T_C < t)P(T_C < T_{-1}) + P(T_{-1} + T < t)P(T_{-1} < T_C)$$
(C.4)

After differentiating with respect to t, eq C.4 becomes

$$f_{T_2}(t) = f_{T_C}(t)P(t < T_{-1}) + \int_0^t dt_1 f(t - t_1)f_{T_{-1}}(t_1)P(t_1 < T_C)$$
(C.5)

From the expression for $f_{T_{-1}}(t)$, $P(t \le T_{-1})$ is given by $\exp(-k_{-1}t)$, while $P(t_1 \le T_C)$ is given by $1 - \int_0^{t_1} dt_2 f_{T_C}(t_2)$. Hence the Laplace transform of eq C.5 is

$$\hat{f}_{T_2}(s) = \hat{f}_{T_c}(s+k_{-1}) + \frac{k_{-1}}{s+k_{-1}}\hat{f}(s)[1-\hat{f}_{T_c}(s+k_{-1})] \quad (C.6)$$

Substituting eq C.6 into eq C.3 and rearranging, we find

$$\hat{f}(s) = \hat{f}_{T_{c}}(s + k_{-1}) / \left(\frac{k_{1}[S] + s}{k_{1}[S]} - \frac{k_{-1}}{k_{-1} + s} [1 - \hat{f}_{T_{c}}(s + k_{-1})] \right)$$
(C.7)

which is the expression shown in eq 32.

Note Added after ASAP Publication. The Note Added in Proof was included with this paper on September 22, 2005. The

article was published ASAP on 8/2/05. The appended version was reposted on 9/23/05.

References and Notes

(1) Michaelis, L.; Menten, M. L. Biochem. Z. 1913, 49, 333.

- (2) Funatsu, T.; Harada, Y.; Tokunaga, M.; Saito, K.; Yanagida, T. Nature 1995, 374, 555.
- (3) Noji, H.; Yasuda, R.; Yoshida, M.; Kinosita, K. Nature 1997, 386, 299.
 - (4) Lu, H. P.; Xun, L.; Xie, X. S. Science 1998, 282, 1877.

(5) Ha, T.; Ting, A. Y.; Liang, J.; Caldwell, W. B.; Deniz, A. A.; Chemla, D. S.; Schultz, P. G.; Weiss, S. Proc. Natl. Acad. Sci. U.S.A. 1999, 96, 893.

(6) Edman, L.; Rigler, R. *Proc. Natl. Acad. Sci. U.S.A.* 2000, *97*, 8266.
(7) Yasuda, R.; Noji, H.; Yoshida, M.; Kinosita, K.; Itoh, H. *Nature* 2001, *410*, 898.

(8) Zhuang, X.; Kim, H.; Pereira, M. J. B.; Babcock, H. P.; Walter, N. G.; Chu, S. *Science* **2002**, *296*, 1473.

(9) Yildiz, A.; Forkey, J. N.; McKinney, S. A.; Ha, T.; Goldman, Y. E.; Selvin, P. R. *Science* **2003**, *300*, 2061.

(10) Bustamante, C.; Bryant, Z.; Smith, S. B. *Nature* 2003, *421*, 423.
(11) van Oijen, A. M.; Blainey, P. C.; Crampton, D. J.; Richardson, C. C.; Ellenberger, T.; Xie, X. S. *Science* 2003, *301*, 1235.

- (12) Yang, H.; Luo, G.; Karnchanaphanurach, P.; Louie, T.-M.; Rech, I.; Cova, S.; Xun, L.; Xie, X. S. *Science* **2003**, *302*, 262.
- (13) Tang, J.; Mei, E.; Green, C.; Kaplan, J.; DeGrado, W. F.; Smith, A. B., III; Hochstrasser, R. M. J. Phys. Chem. B 2004, 108, 15910.

(14) Velonia, K.; Flomenbom, O.; Loos, D.; Masuo, S.; Cotlet, M.; Engelborghs, Y.; Hofkens, J.; Rowan, A. E.; Klafter, J.; Nolte, R. J. M.; de Schryver, F. C. Angew. Chem. Int. Ed. **2005**, *44*, 560.

(15) Flomenbom, O.; Velonia, K.; Loos, D.; Masuo, S.; Cotlet, M.; Engelborghs, Y.; Hofkens, J.; Rowan, A. E.; Nolte, R. J. M.; van der Auweraer, M.; de Schryver, F. C.; Klafter, J. *Proc. Natl. Acad. Sci. U.S.A.* **2005**, *102*, 2368.

(16) English, B. P.; Min, W.; van Oijen, A. M.; Lee, K. T.; Luo, G.;
Sun, H.; Cherayil, B. J.; Kou, S. C.; Xie, X. S. Manuscript in preparation.
(17) Xie, X. S. Single Molecule 2001, 4, 229.

(18) Frauenfelder, H.; Sligar, S. G.; Wolynes, P. G. Science **1991**, 254, 1598. Austin, R.; Beeson, K. W.; Eisenstein, L.; Frauenfelder, H. Biochemistry **1975**, 14, 5355.

- (19) Agmon, N.; Hopfield, J. J. J. Chem. Phys. 1983, 79, 2042.
- (20) Zwanzig, R. Acc. Chem. Res. 1990, 23, 148.
- (21) Zwanzig, R. J. Chem. Phys. 1992, 97, 3587.
- (22) Gehlen, J.; Marchi, M.; Chandler, D. Science 1994, 263, 499.
- (23) Wang, J.; Wolynes, P. Phys. Rev. Lett. 1995, 74, 4317.
- (24) Eizenberg, N.; Klafter, J. J. Chem. Phys. 1996, 104, 6796.
- (25) Bicout, D. J.; Szabo, A. J. Chem. Phys. 1998, 108, 5491.
- (26) Schenter, G. K.; Lu, H. P.; Xie, X. S. J. Phys. Chem. A 1999, 103, 10477.
- (27) Karplus, M. J. Phys. Chem. B 2000, 104, 11.
- (28) Yang, S.; Cao, J. S. J. Chem. Phys. 2002, 117, 10996.
- (29) Qian, H.; Elson, E. L. Biophys. Chem. 2002, 101, 565.
- (30) Xie, X. S. J. Chem. Phys. **2002**, 117, 11024.
- (31) Barsegov, V.; Shapir, Y.; Mukamel, S. Phys. Rev. E 2003, 68, 011101.
 - (32) Brown, F. L. H. Phys. Rev. Lett. 2003, 90, 028302.
 - (33) Kou, S. C.; Xie, X. S. Phys. Rev. Lett. 2004, 93, 180603.
- (34) Min, W.; Luo, G.; Cherayil, B. J.; Kou, S. C.; Xie, X. S. Phys. Rev. Lett. 2005, 94, 198302.
- (35) Asbury, C. L.; Fehr, A. N.; Block, S. M. Science 2003, 302, 2130.
 (36) Schnitzer, M. J.; Block, S. M. Cold Spring Harbor Symp. Quant. Biol. 1995, 60, 793.
- (37) Svoboda, K.; Mitra, P. P.; Block, S. M. Proc. Natl. Acad. Sci. U.S.A. 1994, 91, 11782.
- (38) Segel, I. H. Enzyme Kinetics: Behavior and Analysis of Rapid Equilibrium and Steady-State Enzyme Systems; Wiley: New York, 1993.
- (39) Yang, S.; Cao, J. S. J. Phys. Chem. B **2001**, 105, 6536.
- (40) Ninio, J. Proc. Natl. Acad. Sci. U.S.A. 1987, 84, 663.
- (41) Flomenbom, O.; Klafter, J.; Szabo, A. Biophys. J. 2005, 88, 3780.
- (42) Schnitzer, M. J.; Block, S. M. Nature 1997, 388, 386.